

## Format for ANSWERING REVIEWERS



December 30, 2015

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 22982-review.doc).

**Title:** Biopsy of parotid masses: Review of current techniques

**Author:** Sananda Haldar, Joseph D Sinnott, Kemal M Tekeli, Samuel S Turner, David C Howlett

**Name of Journal:** *World Journal of Radiology*

**ESPS Manuscript NO:** 22982

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2 Revision has been made according to the suggestions of the reviewer

We have added an abstract and added some typical images as suggested. We hope you find these interesting and to your approval.

We have also mentioned CT in the introduction in the paragraph which also mentions MR. We have reviewed the format and hope it fits with the rules of WJR.

3 References and typesetting were corrected

Thank you again for publishing our manuscript in the *World Journal of Radiology*.

Regards,

A handwritten signature in black ink, appearing to read 'J. Sinnott'.

Mr Joseph Sinnott  
Core Surgical Training  
East Sussex Healthcare Trust  
UK  
j.d.sinnott@gmail.com